Author's response to reviews

Title: Influence of the incremental step size in work rate on exercise response and gas exchange in patients with pulmonary hypertension

Authors:

Sven Glaeser (glaeser.sven@gmail.com)
Sven Lodziewski (sven.lodziewski@uni-greifswald.de)
Beate Koch (kochb@uni-greifswald.de)
Christian F. Opitz (opitz@web.de)
Henry Voelke (voelcke@uni-greifswald.de)
Ralf Ewert (ewert@uni-greifswald.de)

Version: 4 Date: 18 February 2008

Author's response to reviews:

Dear editor, please find enclosed the requested corrections:

- cover sheet with corresponding author and e-mails addresses was added
- conclusion section was added
- authors contributions was changed to authors' contributions
- no acknowledgements were applied
- reference list was moved into the main text
- tables were changed in the requested way
- table style was changed
- figures were saved as tiff files, legends were added at the end of the main document
- figures were cropped

Thank you very much in advance,

Sven Glaeser, M.D.